Author’s response to reviews

Title: Is Family-Integrated-Care in Neonatal Intensive Care Units feasible and good to preterm infants in China: Study protocol for a cluster randomized controlled trial

Authors:
Mingyan Hei (2533749531@qq.com)
Xiangyu Gao (g.xy@163.com)
Xirong Gao (gaoxirong@126.com)
Shaohan Nong (nongsh98@hotmail.com)
Aimin Zhang (lilly610@sina.com)
Qianshen Zhang (zhangqianshen2005@163.com)
Shoo Lee (sklee@mtsinaion.ca)

Version: 1 Date: 25 Dec 2015

Author’s response to reviews:

Dec 25, 2015

Dear Editor,

Trials

Thank you for informing us about the minor revision of the manuscript “Is Family-Integrated-Care in Neonatal Intensive Care Units feasible and good to preterm infants in China: Study protocol for a cluster randomized controlled trial” (No. TRLS-D-15-00133), which was submitted to Trials on August 28, 2015.

Here goes the point-to-point reply to the comments of the reviewers:
Please ensure the title conforms to journal style for study protocol articles. The title should follow the format “____________: study protocol for a randomized controlled trial.” Please note that the title in the submission system should match that of your manuscript.

Reply: Yes, the title has been reformatted to “Is Family-Integrated-Care in Neonatal Intensive Care Units feasible and good for preterm infants in China: Study protocol for a cluster randomized controlled trial”.

Please include the email address of each author on the title page.

Reply: The email address of each author has been added on the title page following each author’s affiliation in a bracket.

Please include the date your study was registered with your trial registration number at the end of the Abstract.

Reply: The date of the study registration with the trial registration number has been added at the end of the Abstract as “Trial registration: ChiCTR-TRC-14004736 (June 2, 2014)”.

Please change the title 'Introduction' to 'Background'.

Reply: The title “Introduction” has been changed into “Background”.

Please include a statement in your Methods section explaining that you obtained informed consent from each participant.

Reply: A statement has been added at the end of the 1st paragraph of “Ethical approval and trial registration” in the section of Method as “The study got the ethical approval from Ethics Committee ….. is ChiCTR-TRC-14004736. An informed consent has been obtained from each participant.”.

Please include a figure title and legend section after the reference list. The figures should not be included in the main body of the manuscript.
Reply: A figure title and legend section has been added in a new section after the reference list. We have made sure that the figure was not included in the main body of the manuscript.

Please be informed that all revisions made have been highlighted in red letters.

Thank you very much for your hard work and attention. Looking forward to hearing from you for the accepting of the manuscript!

Yours sincerely,

Dr. Mingyan Hei

The corresponding author

Email: 2533749531@qq.com